

## REVIEW HISTORY

**Manuscript Title:** Constructing Credibility and Narrative Truth in Southern African Childhood Memoirs

Submitted: 20 April, 2026

Accepted: 26 May, 2026

Published: 31 May, 2026

Corresponding Author: **Malowa Phelphone Malowa**

Email: [phelphone@gmail.com](mailto:phelphone@gmail.com)

Doi: <https://dx.doi.org/10.19044/llc.v13no1a2>

### **PEER REVIEW**

Reviewer 1: Charles Omondi Odondo

Reviewer 2: George Ochieng Abidha